Author's response to reviews

Title: Which Criteria should be used to define Type 2 Diabetes Remission after Bariatric Surgery?

Authors:

Ana M Ramos-Levi (ana_ramoslevi@hotmail.com)
Lucio Cabrerizo (lcabrerizo.hcsc@salud.madrid.org)
Pilar Matia (mmatia.hcsc@salud.madrid.org)
Andres Sanchez-Pernaute (pernaute@yahoo.com)
Antonio J Torres (ajtorresgarcia@gmail.com)
Miguel A Rubio (marubioh@gmail.com)

Version: 2 Date: 19 September 2012

Author's response to reviews:

Following the instructions of the Journal Editorial Office, we have added the information requested:

1. We have specified that research was performed with the approval of our local Ethics Committee.
2. We have clarified that for our retrospective study, written informed consent was attained from all patients.

We hope that these changes satisfy the requirements for the article to be considered for revision and publication.